

The Unequal Geographic Burden of Federal Taxation Author(s): David Albouy Source: *Journal of Political Economy*, Vol. 117, No. 4 (August 2009), pp. 635-667 Published by: <u>The University of Chicago Press</u> Stable URL: <u>http://www.jstor.org/stable/10.1086/605309</u> Accessed: 07/01/2014 13:26

Your use of the JSTOR archive indicates your acceptance of the Terms & Conditions of Use, available at http://www.jstor.org/page/info/about/policies/terms.jsp

JSTOR is a not-for-profit service that helps scholars, researchers, and students discover, use, and build upon a wide range of content in a trusted digital archive. We use information technology and tools to increase productivity and facilitate new forms of scholarship. For more information about JSTOR, please contact support@jstor.org.



The University of Chicago Press is collaborating with JSTOR to digitize, preserve and extend access to Journal of Political Economy.

http://www.jstor.org

The Unequal Geographic Burden of Federal **Taxation**

David Albouy

University of Michigan and National Bureau of Economic Research

In the United States, workers in cities offering above-average wagescities with high productivity, low quality of life, or inefficient housing sectors-pay 27 percent more in federal taxes than otherwise identical workers in cities offering below-average wages. According to simulation results, taxes lower long-run employment levels in high-wage areas by 13 percent and land and housing prices by 21 and 5 percent, causing locational inefficiencies costing 0.23 percent of income, or \$28 billion in 2008. Employment is shifted from north to south and from urban to rural areas. Tax deductions index taxes partially to local cost of living, improving locational efficiency.

I. Introduction

Wage and cost-of-living levels vary considerably across cities in the United States, yet the federal tax code does not take this variation into

I thank Alan Auerbach, Timothy Bartik, David Card, Allan Collard-Wexler, Tom Davidoff, Gilles Duranton, Rob Gillezeau, Erica Greulich, Jim Hines, Darren Lubotsky, Bert Lue, Erin Metcalf, Enrico Moretti, John Quigley, Marit Rehavi, Emmanuel Saez, and William Strange; participants of seminars at Chicago-Harris, Michigan, Northwestern, New York University, the Philadelphia Federal Reserve Bank, Public Policy Institute of California, Syracuse, Toronto-Rotman, University of California, Berkeley, University of California, San Diego, University of Texas at Austin, and Wisconsin; and presentations at the 2007 NBER Summer Institute on Taxation, 2007 National Tax Association Annual Meetings, and 2009 Centre for Economic Policy and Research Public Policy Symposium for their help, input, and advice. Furthermore, I am grateful to the Burch Center for Tax Policy and Public Finance and the Fisher Center for Real Estate and Urban Economics for their financial assistance, as well as the National Tax Association, the W. E. Upjohn Institute for Employment Research, and the American Real Estate and Urban Economics Association for generously awarding my dissertation, "Causes and Consequences of Unequal Federal Tax-ation and Spending across Regions," which this article is based on. Any mistakes are my own.

[Journal of Political Economy, 2009, vol. 117, no. 4] © 2009 by The University of Chicago. All rights reserved. 0022-3808/2009/11704-0002\$10.00

account. Since federal taxes are based on nominal incomes, workers with the same real income pay higher taxes in high-cost areas than in low-cost areas, without receiving additional benefits. Recognizing this, the Tax Foundation (Dubay 2006, 1) argues, "the nation is not only redistributing income from the prosperous to the poor, but from the middle-income residents of high-cost states to the middle-income residents of low-cost states." While the Tax Foundation has suggested a flat tax to remedy this problem (Hoffman and Moody 2003), politicians from high-cost areas have proposed indexing federal taxes and benefits to local costs, arguing that workers with the same real incomes should pay the same nominal taxes.

For federal taxes to not distort the location choices of workers, the correct principle is that taxes should be independent of where workers live so that location-wise they are effectively lump sum. The current system taxes workers more for taking jobs in higher-paying cities, blunting the incentive to live in these cities, characterized by high firm productivity and low quality of life. For example, in the New York metropolitan area, wage levels are 21 percent above the national average, which, interacted with an effective marginal tax rate of 33 percent, creates a 7 percent federal surtax on labor income for locating there. Unlike local tax differences, federal tax differences of this kind are not compensated with higher levels of local spending and may therefore affect location choices substantially.

Because federal taxes are not indexed to local wage levels, workers are induced to leave cities with high wages and move to cities with low wages. As a result, unindexed federal taxes lower employment levels and property values in high-wage cities while having the opposite effect on low-wage cities. In equilibrium, these price changes compensate workers for federal tax differences across cities, but the resulting geographic distribution of employment is inefficient, reducing overall welfare.

The unequal distribution of federal taxes that results from wage differences across cities does not depend on the progressivity of taxes and cannot be eliminated with a flat tax. The view that workers with the same real incomes should pay the same nominal taxes holds true across cities that vary in the productivity of their firms, as nominal incomes merely track cost-of-living differences across these cities. However, this view is incorrect across cities that vary in quality of life, as nicer cities have a higher cost of living but lower nominal wage levels and hence a lower federal tax burden. Indexing the tax code to local costs would eliminate federal tax differences across cities that vary in productivity but exacerbate them across cities that vary in quality of life.

An empirical simulation for the United States, below, reveals that workers with the same skills pay up to 27 percent more in federal taxes in high-wage cities than in low-wage cities. The federal government

effectively taxes workers for living in large cities while subsidizing them to live in rural areas. Taxes also fall more heavily on the Northeast, Pacific, and Great Lakes regions and less on the South. Controlling for socioeconomic disparities, approximately \$270 billion each year are transferred horizontally from high-wage areas to low-wage areas. These findings partly confirm Senator Patrick Moynihan's (2000) claims in 24 years of reports, entitled *The Federal Budget and the States*, that the "federal balance of payments" across areas is highly unequal, although these reports do not control for socioeconomic differences across regions, nor do they consider the effects on local employment or prices.

Journalist Malcolm Gladwell (1996) writes that the inequality in the federal balance of payments "is according to urban experts and economists one of the best-kept secrets in American politics" and that "the decline of many northeastern American cities may be due not just to mismanagement—as is now popularly imagined—but to the emptying of their coffers by the federal government." Such a view is supported by the simulation: over the long run, federal taxes have lowered employment, housing prices, and land values in high-wage areas by 13, 5, and 21 percent, respectively, and done the opposite in low-wage areas. Overall, federal taxes have tilted the geographic distribution of employment away from the North toward the South and away from urban areas toward rural areas, creating a welfare loss estimated at 0.23 percent of income, or \$28 billion in 2008. Without federal tax deductions for mortgage interest and local taxes, this loss would be even larger.

Previous research about how federal taxes interact with local prices contains some important findings but has been too narrow or informal to guide policy comprehensively. Wildasin (1980) finds that federal taxes on labor income cause mobile workers to locate inefficiently across cities offering different wages but focuses on conditions characterizing efficiency rather than the results of inefficiency. Without referring specifically to taxation, Glaeser (1998) argues that federal transfer levels should not be tied to local price levels, as this implicitly subsidizes recipients to live in expensive, high-quality-of-life cities. More generally, Kaplow (1995) and Knoll and Griffith (2003) also allow productivity differences to affect local wages and prices, leading them to consider the benefits of indexing taxes to local wages. Although insightful, their informal arguments leave open the exact consequences of failing to index the tax code, raising the need for more rigorous quantitative analysis.¹

¹Kaplow's (1995) analysis holds prices fixed and presents an index formula that does not equalize nominal tax payments across areas. Knoll and Griffith (2003) assume that a flat tax on income does not change prices or reallocate resources; this assumption, as shown below, does not hold in general equilibrium. Other work considers how tax deductions interact with local prices. Research by Gyourko and Sinai (2003, 2004) and Brady,

Section II introduces a model of mobile workers who live in cities with attributes that generate differences in costs of living, wages, and federal tax burdens. Section III describes the federal tax differences that arise in equilibrium and how they affect local prices. Section IV examines how taxes distort location decisions and how to calculate the resulting efficiency loss. Then, Section V considers the effect of indexing taxes to local wages or costs of living and demonstrates how tax deductions for locally produced goods, such as housing, produce a mild form of cost indexation. Section VI calibrates the model and simulates how differential taxes affect the distribution of local prices, employment, and welfare, taking into account differential federal spending patterns.

II. Theoretical Setup

To explain why prices and tax burdens differ across cities, I adapt the general-equilibrium model of Rosen (1979) and Roback (1980, 1982), incorporating federal taxes. The national economy is closed and contains many cities, indexed by *j*, which trade with each other and share a homogeneous population of mobile workers. These workers consume a numeraire traded good, x, and a nontraded "home" good, y, with local price p^{i} . Cities differ in three types of exogenous attributes. Quality of life, Q^{j} , may be affected by amenities such as weather or safety. Productivity in the traded-good sector, A_x^j (or "trade productivity"), may be due to natural advantages, such as a harbor, or to agglomeration economies, such as input sharing. Productivity in the home-good sector, A_{Y}^{j} (or "home productivity"), may be affected by natural advantages or regulations affecting residential housing. The average value of each attribute is set to one. Although some city attributes may indeed be endogenous, it is safe to consider them exogenous if federal taxes do not significantly affect their relative levels across cities.

Firms produce traded and home goods out of land, capital, and labor. Factors receive the same payment in either sector. Land, L, is fixed in supply in each city at L^j and is paid a city-specific price, r^j . Capital, K, is fully mobile and is paid the price i everywhere. The supply of capital in each city is denoted K^j , with the aggregate level of capital fixed at K^{TOT} ; thus $\sum_j K^j = K^{\text{TOT}}$. Labor, N, is also fully mobile, but because workers care about local prices and quality of life, wages, w^j , may vary across cities. Workers have identical tastes and endowments, and each

Cronin, and Houser (2003) tabulates how mortgage and local-tax deductions disproportionately benefit high-cost areas but neglects how these deductions may offset the unequal burden of federal taxes from wage differences. Surveys of the possible benefits of tax deductions for mortgage interest (e.g., Glaeser and Shapiro 2003) or local taxes (e.g., Kaplow 1996) do not consider their interurban locational effects.

supplies a single unit of labor. The total number of workers is fixed at N^{TOT} , so $\sum_j N^j = N^{\text{TOT}}$. Workers own identical diversified portfolios of land and capital, which pay an income $R = (1/N^{\text{TOT}}) \sum_j r^j L^j$ from land and $I = \overline{i} (K^{\text{TOT}}/N^{\text{TOT}})$ from capital. Total income $m^j \equiv R + I + w^j$ varies across cities only as wages vary. Out of this income, workers pay a federal income tax of $\tau(m^j)$. Deductions are introduced in Section V.²

Workers' preferences are modeled by a utility function, U(x, y; Q), that is quasi-concave and homothetic over x and y and increasing in Q. The corresponding expenditure function is $e[p, u, \tau(m); Q] \equiv \min_{x,y}[x + py + \tau(m) : U(x, y; Q) \ge u]$; Q is assumed to enter neutrally into the utility function and is normalized so that $e[p, u, \tau(m); Q] = [e(p, u) + \tau(m)]/Q$, where $e(p, u) \equiv e(p, u, 0; 1)$. Since workers are fully mobile, their utility must be the same across all inhabited cities so that higher prices, lower quality of life, or higher taxes must be compensated with greater income:

$$[e(p^{j}, \bar{u}) + \tau(m^{j})]/Q^{j} = m^{j};$$
(1)

 \bar{u} is the level of utility attained by all workers, regardless of each worker's federal tax burden.³

Operating under perfect competition, firms produce traded and home goods according to the functions $X = A_x F_x(L_x, N_x, K_x)$ and $Y = A_y F_y(L_y, N_y, K_y)$, where F_x and F_y are concave and exhibit constant returns to scale.⁴ Unit cost in the traded-good sector is $c_x(r, w, i)/A_x \equiv$ min $_{L,N,K}[rL + wN + iK : A_x F(L, N, K) = 1]$. A symmetric definition holds for unit cost in the home-good sector, c_y . All factors are fully employed: $L_x^j + L_y^j = L^j, N_x^j + N_y^j = N^j$, and $K_x^j + K_y^j = K^j$. As markets are competitive, firms make zero profits in equilibrium so that for given output prices, more productive cities pay higher rents and wages, and the following conditions hold in all cities j where production occurs:

$$c_X(r^j, w^j, \bar{i})/A_X^j = 1,$$
 (2)

$$c_{Y}(r^{j}, w^{j}, \bar{i})/A_{Y}^{j} = p^{j}.$$
 (3)

² Because markets are perfectly competitive, the economic incidence is unchanged if the nominal incidence of taxes is placed on firms' labor costs rather than on workers' wage incomes. Consumption taxes in this model are equivalent to income taxes; taxes on production are largely equivalent, except for the portion that falls on capital and land.

⁴ Non-Hicks-neutral productivity differences have similar impacts on relative prices across cities but not on relative quantities.

³ The model generalizes to a case with workers that supplies different fixed amounts of labor if these workers are perfect substitutes in production, have identical homothetic preferences, and earn equal shares of income from labor. More general types of worker heterogeneity are considered in Albouy (2008*b*), including the case in which some workers are immobile or differ in their attachment to particular cities, simulating the effects of moving costs. This explains how federal tax changes can have redistributive effects across areas when tastes are heterogeneous or moving costs are substantial.

This analysis models a single federal government that collects tax revenues, makes transfers, and uses the net balance to buy traded goods that are transformed into a federal public good, such as defense. This federal public good benefits workers everywhere equally, and its level is held fixed. Federal taxes are modeled net of federal transfers. Naturally, federal means-tested benefits increase the effective marginal tax rate for some workers.⁵ In addition, it matters if federal tax payments are tied to federal transfers. In the United States, workers in high-wage areas pay more in payroll taxes and then receive higher social security benefits later in life. Thus, the marginal benefit of paying these taxes should be subtracted from the effective marginal income tax rate.

The local public sector does not need to be modeled explicitly. If local government provides goods efficiently, as in the Tiebout (1956) model, these goods can be treated as consumption goods. Furthermore, efficiency differences across local public sectors may be subsumed into differences in Q^j (Gyourko and Tracy 1989, 1991) or A_Y^j . Taxes levied at the subnational level can also be distributed unequally across areas when wages vary within a subnational jurisdiction, such as a state. State taxes are incorporated into the simulation below, in which their effects are small; for expositional ease, they are ignored here.

For workers, denote the expenditure shares of traded goods, home goods, and taxes as $s_x^j \equiv x^{j/m^j}$, $s_y^j \equiv p^j y^{j/m^j}$, and $s_T^j = \tau(m^j)/m^j$; denote the shares of income received from land, labor, and capital income as $s_R^j \equiv R/m^j$, $s_w^j \equiv w^{j/m^j}$, and $s_I^j \equiv I/m^j$. For firms, denote the cost shares of land, labor, and capital in the traded-good sector as $\theta_L^j \equiv r^j L_x^j/X^j$, $\theta_N^j \equiv w^j N_X^j/X^j$, and $\theta_K^j \equiv \bar{i} K_N^j/X^j$; denote similar cost shares in the home-good sector as ϕ_L^j , ϕ_N^j , and ϕ_K^j . Assume, as is likely, that home goods are more cost intensive in land relative to labor than traded goods, that is, $\phi_L^j/\phi_N^j > \theta_L^j/\theta_N^j$.

III. Price and Federal Tax Differences across Cities

Federal taxes on labor income affect how prices vary cross-sectionally across cities with different attributes. To analyze this, assume that there are enough cities varying in the three city attributes, Q, A_x , and A_y , to treat these attributes as continuous variables. The equilibrium condi-

⁵ This is complicated by eligibility requirements for programs that vary by state or county. Furthermore, some benefit levels are tied to local prices, such as housing programs, although these programs tend to be small. Inasmuch as they are valued, local goods provided by the federal government may be treated as transfers, as can intergovernmental transfers that increase the supply of local government goods. It should be noted that federal matching rates for many programs (e.g., Medicaid) decline with average state income. The complicated nature of these transfers makes it useful to consider some federal transfers separately from an overall tax schedule, as in Sec. VI.D.

tions (1), (2), and (3) implicitly define the prices w^j , r^j , and p^j —and the federal tax, $\tau(m^j)$, which depends on them—as a function of Q^j , A_X^j , and A_Y^j . These conditions may be log-linearized to express a particular city's price differentials in terms of its city-attribute differentials, each relative to the national average. These differentials are expressed in logarithms so that, for any variable z, $\hat{z}^j = \ln z^j - \ln \bar{z} \approx (z^j - \bar{z})/\bar{z}$ approximates the percentage difference in city j of z relative to the geometric average \bar{z} . Values in the presence of income taxes are not subscripted; counterfactual values under a uniform, utility-equivalent, lump-sum tax are subscripted by zero, for example, \hat{z}_0^j . The change in z due to income taxes is denoted with a d, so $dz^j = z^j - z_0^j$ and $d\hat{z}^j =$ $\hat{z}^j - \hat{z}_0^j$. In an average city, $\hat{z}^j = \hat{z}_0^j = d\hat{z}^j = 0$.

Log-linearized versions of (1), (2), and (3) describe how prices covary with city attributes:

$$s_w \hat{w}^j - s_y \hat{p}^j = \tau' s_w \hat{w}^j - \hat{Q}^j, \tag{4a}$$

$$\theta_L \hat{r}^j + \theta_N \hat{w}^j = \hat{A}_X^j, \tag{4b}$$

$$\boldsymbol{\phi}_L \hat{\boldsymbol{r}}^j + \boldsymbol{\phi}_N \hat{\boldsymbol{w}}^j - \hat{\boldsymbol{p}}^j = \hat{A}^j_Y.$$
(4c)

These equations are first-order approximations around a nationally representative city, so the share values, without superscripts, are national averages. Equation (4a) states how before-tax real income, given by the nominal income difference, $s_w \hat{w}^j$, net of the cost-of-living difference, $s_{i}\hat{p}^{j}$, compensates for lower quality of life, $-\hat{Q}^{j}$, and higher federal taxes, $\tau' s_{m} \hat{w}^{j}$. This last term is the income tax differential as a fraction of total income, $\tau' s_w \hat{w}^j = \tau' \hat{m}^j \equiv d\tau^{j/m}$, due to the wage differential \hat{w}^j . For example, if a city offers 10 percent higher wages, the share of income from wages is 75 percent, and the marginal tax rate is 33 percent, then workers of the city pay additional taxes equal to 2.5 percent of income. The effects of a federal tax differential are similar to that of a head tax on workers for living in city j, except that the federal tax differential depends on an endogenous wage differential, \hat{w}^{j} , rather than being set exogenously. Equations (4b) and (4c) demonstrate how high productivity in each sector results in high factor prices relative to the output price in equilibrium.

The tax differentials depend on the wage differentials, which may be written

$$\hat{w}^{j} = \hat{w}_{0}^{j} + \underbrace{\frac{\theta_{L}}{\theta_{N}} \frac{1}{s_{R}} \tau' s_{w} \hat{w}^{j}}_{d\hat{w}^{j}} = \frac{1}{1 - (\theta_{L}/\theta_{N})(s_{w}/s_{R})\tau'} \hat{w}_{0}^{j},$$
(5)

where the wage differential under a neutral, utility-equivalent, lumpsum tax,

$$\hat{w}_0^j = \frac{1}{\theta_N s_R} (s_y \phi_L \hat{A}_X^j - \theta_L \hat{Q}^j - s_y \theta_L \hat{A}_Y^j), \tag{6}$$

relates how wages rise with trade productivity and fall with quality of life or home productivity. The first equality of (5) demonstrates that firms paying a positive wage differential without income taxes, \hat{w}_0^i , pay an additional wage differential, $d\hat{w}^i$, to help compensate for higher income taxes. The term multiplying \hat{w}_0^i after the second equality exceeds one, meaning that income taxes increase wage differences across cities.⁶

Combining equations $d\tau^{j}/m = \tau' s_{w} \hat{w}^{j}$, (5), and (6), the tax differential in terms of city attributes is

$$\frac{d\tau^{j}}{m} = \tau' \frac{1}{1 - (\theta_{L}/\theta_{N})(s_{w}/s_{R})\tau'} \frac{s_{w}}{\theta_{N}s_{R}} \left(s_{y}\phi_{L}\hat{A}_{X}^{j} - \theta_{L}\hat{Q}^{j} - s_{y}\theta_{L}\hat{A}_{Y}^{j}\right).$$
(7)

As do wages, federal taxes rise with trade productivity and fall with quality of life or home productivity. Spatially, the income tax operates as if the federal government supplemented a uniform lump-sum tax with a revenue-neutral system of head taxes, which vary across cities according to equation (7).

Land rent and home-good price differentials can be decomposed similarly:

$$\hat{r}^{j} = \hat{r}_{0}^{j} - \underbrace{\frac{1}{s_{R}} \frac{d\tau^{j}}{m}}_{d\hat{r}^{j}},$$
(8a)

$$\hat{p}^{j} = \hat{p}_{0}^{j} - \underbrace{\left(\phi_{L} - \frac{\theta_{L}}{\theta_{N}} \phi_{N} \right)}_{d\hat{\mu}^{j}} \frac{1}{s_{R}} \frac{d\tau^{j}}{m}, \qquad (8b)$$

where the rent and price differentials under a utility-equivalent lumpsum tax are

$$\hat{r}_{0}^{j} = \frac{1}{s_{R}} (\hat{Q}^{j} + s_{x} \hat{A}_{x}^{j} + s_{y} \hat{A}_{y}^{j}), \qquad (9a)$$

⁶ The solution requires the identities $s_R = (s_x + s_T)\theta_L + s_y\phi_L$ and $s_w = (s_x + s_T)\theta_N + s_y\phi_N$. Expressions for price differentials without taxation equivalent to (6), (9a), and (9b) are found in Roback (1980). Those expressions are not log-linearized and ignore nonlabor income and the accounting identities. Gyourko and Tracy (1989) develop expressions similar to (5) and (8a) for wage and rent changes in the presence of local income taxes in the simpler case where $\phi_L = 1$. Their expressions look very different, as they are not log-linearized or simplified in the same way. These analyses do not refer to federal taxes or deductions.

$$\hat{p}_0^j = \frac{1}{\theta_N s_R} [(\theta_N \phi_L - \theta_L) \phi_N \hat{Q}^j + \phi_L s_w \hat{A}_X^j - \theta_L s_w \hat{A}_Y^j].$$
(9b)

Both land rents and home-good prices increase with quality of life and trade productivity, although land rents rise and home-good prices fall with home productivity. Equation (8a) reveals how additional federal taxes are fully capitalized into land rents as $s_R \cdot m \cdot d\hat{r}^j = -d\tau^j$, which implies $dr^j \cdot L^j = -N^j \cdot d\tau^{j,7}$ Equation (8b) reveals how taxes are capitalized into the price of home goods, depending on their land intensity. Overall, taxes lower relative land and home-good prices in cities with higher trade productivity, lower quality of life, or lower home productivity.⁸

Workers are compensated for higher taxes through a combination of higher wages and lower home-good prices. Using the expression for $d\hat{w}^{j}$ in equation (5), it is possible to show that the fraction of taxes compensated through wages, $dw^{j}/d\tau^{j}$, equals λ_{L}/λ_{N} , denoting the ratio of the fraction of land in the traded goods sector, $\lambda_{L} \equiv (1 - s_{y})\theta_{L}/s_{R}$, to the fraction of labor in the traded sector, $\lambda_{N} = (1 - s_{y})\theta_{N}/s_{w}$. The less land is used in traded-good production, the less total costs fall when taxes cause land rents to fall, and thus the less wages increase and the more lower land rents are passed on to workers through lower homegood prices. This ratio also determines how much quality-of-life advantages are reflected in lower wages rather than higher prices.⁹

The effect of federal taxes on local prices can be shown graphically by assuming that home goods are just land ($\phi_L = 1, A_Y^j = 1$), so that p = r, and that, initially, workers everywhere pay a uniform lump-sum tax of *T*. Figure 1 illustrates the case of a highly trade-productive city, say Chicago (labeled *C*), and an average city, say Nashville, with pro-

⁷ If land is not shared equally across the population, increases in the marginal (but not average) tax rate benefit land owners in low-wage cities and hurt those in high-wage cities. Utilities cease to be equal across workers, but this does not change the resulting equilibrium if preferences are homothetic. As home goods consist mainly of durable housing, supply of home goods could take time to adjust to this equilibrium in response to a tax change. In the short run, the housing supply is relatively fixed. A way to model this is to augment the definition of "land" to include the housing stock and to increase the effective cost shares ϕ_L and θ_L . In the short run, housing price changes are larger and employment changes smaller than in the long run.

⁹ How attributes are capitalized into local prices is discussed in greater detail in Albouy (2009).

⁸ The effect of taxes on prices is sensitive to the assumption that attributes are exogenous. This is most conspicuous with respect to trade productivity, which increases with overall employment because of agglomeration. Higher federal taxes cause employment to fall, lowering trade productivity. This in turn lowers wages, home-good prices, and land rents, magnifying the effects for the latter two while dampening (or possibly reversing) the effect on wages. A simplified example is shown in Albouy (2008*b*). If quality of life falls (rises) with employment, then wage, rent, and price changes are dampened (magnified). If home productivity falls (rises) with employment, then wage and rent effects are dampened (magnified) while price effects are magnified (dampened).



FIG. 1.—Effect of federal taxes on a high trade-productivity city. In a simplified model $(r^i = p^j, Q^j = A_Y^i = 1 \text{ for all } j)$, replacing a lump-sum tax, T, with a utility-equivalent federal income tax, τ , raises wages, w, and lowers rents, r, and employment in Chicago, labeled "*C*," a city with high trade productivity $(A_X^c > 1)$, changing the equilibrium from E_0^c to E^c .

ductivities $A_x^c > 1$ and $\bar{A}_x = 1$. The zero-profit conditions slope downward, as wages must fall as rents rise to keep profits at zero. More productive firms in Chicago pay higher wages or rents, placing its zero-profit condition to the upper right of Nashville's. The worker-mobility condition slopes upward at a rate of *y*, as wages must rise with rents in order for workers to be indifferent between either city. In the tax-free equilibrium, shown at \bar{E} and E_0^c , Chicago is more crowded than Nashville and pays workers a differential, $w_0^c - \bar{w}$, to compensate them for the higher cost of living reflected in $r_0^c - \bar{r}$.

Now replace the lump-sum tax with an income tax set so that workers with an average wage, \bar{w} , pay the same amount of taxes, $\tau(\bar{w} + R + I) = T$, leaving utility unchanged, although now these workers face a positive marginal tax rate, $\tau' > 0$. With this positive marginal tax rate, workers in costlier cities must be paid more before taxes to receive the same compensation after taxes, rotating the mobility condition counterclockwise around its intersection with the horizontal line at \bar{w} , to its slope of $y/(1 - \tau')$. Workers in Chicago at the old equilibrium, E_0^c , are



FIG. 2.—Effect of federal taxes on a high-quality-of-life city. In a simplified model $(r^j = p^j, A_x^j = A_y^j = 1$ for all j), replacing a lump-sum tax, T, with a utility-equivalent federal income tax, τ , lowers wages, w, and raises rents, r, and employment in Miami, labeled "M," a city with high quality of life $(Q^M > 1)$, changing the equilibrium from E_0^M to E^M .

now worse off than in Nashville, as the old compensating differential does not make up for the higher costs and higher taxes. Workers will leave Chicago $(dN^c < 0)$, lowering the demand for land in both production and consumption, causing rents to fall by dr^c and raising the labor-to-land ratio, causing wages to rise by dw^c . At the new equilibrium, E^c , workers are no worse off in Chicago. Firms are no better off, since their cost savings in land are passed off to workers in higher wages. By making Chicago relatively more expensive, the income tax discourages workers from working there, similar to how taxes discourage work by raising the cost of effort relative to leisure.

The case of a city offering a higher quality of life, say Miami, is illustrated in figure 2. Like Chicago, Miami is relatively crowded and has high rents, except that as compensation, workers receive a nicer environment rather than a higher wage. Because land is fixed in supply and used in production, local labor-demand curves are downward sloping; a larger supply of workers in the nicer city lowers the wage. This equilibrium is shown in figure 2, with Nashville and Miami (M) each having qualities of life $\tilde{Q} = 1$ and $Q^M > 1$. Both cities have the same productivity

and so share the same zero-profit condition. Yet, the mobility condition for workers in Miami is located to the lower right, as workers are willing to accept lower wages or pay higher rents to live there. In equilibrium, shown in E_0^M , workers in Miami pay the rent premium $r_0^M - \bar{r}$ and give up the wage differential $w_0^M - \bar{w}$.

Replacing the lump-sum tax with an income tax, workers in Miami pay less tax as they earn below-average wages. A worker is more willing to bid down her wage to live in Miami, as a \$1 reduction in income implies only a $(1 - \tau')$ reduction in consumption. With this effective tax rebate for quality of life, workers in Miami are made better off. Workers are then induced to move to Miami $(dN^M > 0)$ until rents are driven up by dr^M and wages are driven down by dw^M to make Miami no more attractive than other cities. To the extent that higher quality of life is bought through lower pretax wages rather than higher posttax home-good prices, its tax treatment is similar to untaxed fringe benefits: firms located in a city by the beach share tax advantages similar to firms that offer a tax-deductible company car.

The case of a more home-productive city, say Dallas (*D*), may be illustrated simply by assuming $p = r/A_Y^D < r$, as $A_Y^D > \bar{A}_Y = 1$. Lower prices make Dallas workers better off for given wages and rents, shifting the mobility condition to the lower right, as in figure 2. In equilibrium, wages and home-good prices are lower than in Nashville, although rents are higher. Because Dallas workers are paid less, they have lower tax burdens, creating the same tax effects as in Miami.

Taxing labor income may have many advantages, but the burden of income tax is curiously distributed across cities with different attributes. By falling more heavily on cities offering higher wages, federal taxes act like an arbitrary head tax for deciding to live in a city with wage-improving attributes, whatever those attributes may be. The tax is distortionary because workers are artificially attracted to cities that are nicer to live in, more home productive, or less trade productive. At a minimum, it would be preferable to charge an equivalent tax directly on land according to its wage-improving attributes: this would affect land rents in the same way but would not distort location behavior or other prices.¹⁰

¹⁰ If labor supply is elastic, the effect of federal tax differentials cannot be equated directly with head taxes. Real wages fall with quality of life, so if labor supply increases with real wages, labor supply is lower in nicer cities, assuming quality of life and leisure are not substitutes. Thus, in nicer cities, workers will work less and thus avoid taxes even more, increasing the tax advantage that nicer cities have.

IV. Employment Effects and Locational Efficiency

Federal taxes not only influence prices but also cause factors such as labor to move across cities. By making high-wage cities more expensive to live in—or, equivalently, more expensive to hire in—federal taxes induce workers to move away from high-wage areas toward low-wage areas, leading to an efficiency loss from misallocating workers across areas.

The employment effect of a differential tax can be written as

$$d\hat{N}^{j} = \varepsilon \cdot d\tau^{j/m}, \tag{10}$$

where ε is the elasticity of local employment with respect to a local, uncompensated tax, written as a percentage of total income. In principle, reduced-form estimates of this elasticity can be obtained. Furthermore, tax differentials can be obtained directly from data on wages and federal taxes. Thus, employment effects in equation (10) can be calculated without referring to a richer theoretical apparatus. Nevertheless, the theoretical model does imply a structural value for ε . This elasticity is the sum of three long terms, each dependent on a different elasticity of substitution, and is unambiguously negative if $\phi_L/\phi_N > \theta_L/\theta_N$.

Because workers locate in response to federal income taxes, the resulting spatial distribution of employment becomes inefficient, or "locationally inefficient" (Wildasin 1980). I derive the deadweight loss due to this inefficiency by calculating how much revenue the government loses when it replaces a neutral lump-sum tax with an income tax, holding the utility of workers constant. Consistent with Harberger (1964), this deadweight loss, expressed as a fraction of national income, is proportional to half the size of the tax differential times the induced change in migration, averaged across cities.

$$\frac{\text{DWL}}{mN^{\text{TOT}}} = \frac{1}{2} E \left[\frac{d\tau^{j}}{m} d\hat{N}^{j} \right].$$

Whatever the distribution of city attributes, this formula captures the entire efficiency loss from all of the distortions created by unequal geographic taxation, including the indirect distortion on the location of capital. This does assume that city attributes are unaffected by employment levels. Furthermore, as $d\hat{N}^{j} = \varepsilon \cdot d\tau^{j}/m$, the deadweight loss can

be calculated using only data on ε and the variance of income tax differentials:

$$\frac{\mathrm{DWL}}{mN^{\mathrm{TOT}}} = \frac{1}{2} \mathrm{Var} \left(\frac{d\tau^{j}}{m} \right) \varepsilon.$$
(11)

Since $d\tau^{j}/m = \tau' s_{w} \hat{w}^{j}$, the deadweight loss increases with the variance of wage differences across cities.

V. Tax Indexation and Deductions

Since federal taxes make workers locate inefficiently, it is worth considering policies to remedy this problem. Taxes can be indexed to either local wages or local costs: the former is better in theory but arguably harder to implement, whereas the latter oversubsidizes life in nicer locations. If demand for home goods is inelastic, tax deductions for home-good expenditures effectively index taxes partially to local costs.

A. Wage-Level and Cost-of-Living Indexation

Income taxes may be indexed to wages by dividing taxable labor income by the "pay relative" $1 + \hat{w}^j = w^{j/}\bar{w}$, assuming those pay relatives can be correctly measured. With this indexation, a worker's federal taxes do not depend on where she lives, effectively turning the income tax into a neutral lump-sum tax.

Indexing taxes to local cost of living may be easier than indexing taxes to wages, as the prices of homogeneous goods across cities may be easier to measure than the prices of homogeneous units of labor. Presumably, taxes would be indexed to local costs by dividing income by an index $\kappa(p^j)$ —one that ignores quality of life—resulting in taxes $\tau = \tau [m^{j/\kappa}(p^j)]$. An ideal cost-of-living index of this kind is defined in terms of gross expenditures: $\kappa(p^j) = [e(p^j, \bar{u}) + \bar{\tau}]/[e(\bar{p}, \bar{u}) + \bar{\tau}]$, where \bar{p} and $\bar{\tau}$ are the average home-good price and tax burden.

With cost indexation, the tax differential in a city increases with wages and decreases with home-good prices according to the formula $d\tau^{j}/m = \tau'(s_w \hat{w}^j - s_y \hat{p}^j)$. This changes the mobility condition (4a) to

$$s_{j}\hat{p}^{j} - s_{w}\hat{w}^{j} = \frac{\hat{Q}^{j}}{1 - \tau'}.$$
 (12)

With cost-indexed taxes, workers are willing to take a larger fall in pretax real income to improve their quality of life. Substituting equation (12)

into $d\tau^{j}/m = \tau'(s_{w}\hat{w}^{j} - s_{y}\hat{p}^{j})$ reveals that cost-indexed taxes depend only on local quality of life:

$$\frac{d\tau^j}{m} = -\frac{\tau'}{1-\tau'}\hat{Q}^j.$$
(13)

Relative to taxation without indexation, cost indexation eliminates tax differences across cities differing in either type of productivity (A_x or A_y); across these cities, wages rise in step with costs. Thus, indexing with costs is equivalent to indexing with wages. The drawback to cost indexation is that in nicer cities, workers receive two tax advantages: they owe fewer taxes for paying higher prices and for receiving lower wages. The government then massively subsidizes life in nicer cities. While this may sound like a welfare-improving policy, it would actually reduce welfare, as nicer cities would become overcrowded.¹¹

B. Tax Advantages for Housing and Local Taxes

Thus far, I ignored that the federal tax code confers a number of advantages to housing and goods provided by local government. Homeowners benefit from a number of tax advantages in housing consumption, as they are not taxed for the rent they implicitly "pay" themselves when living in their own home and as they can deduct mortgage interest from their income taxes (see Rosen 1985; Poterba 1992). Goods provided by local governments are also subsidized by the federal government, as local and state taxes can be deducted from federal taxes. Since housing and most locally provided government goods, such as education and public safety, are produced locally, these tax advantages may be thought to apply primarily to home goods. Together, these advantages may be modeled by allowing households to deduct a fraction $\delta \in$ [0,1] of home-good expenditures, py, from their federal income taxes so that taxes paid are $\tau(m^j - \delta p^j y)$; δ should be less than one, as these advantages do not apply to certain taxes (e.g., payroll) or to certain home goods, such as haircuts or restaurant meals. Nor are these advantages available to all workers: many renters and homeowners do not itemize deductions for mortgage interest or local taxes.

Incorporating the home-good deduction into the income tax, $\tau(m - \tau)$

¹¹ A handful of U.S. federal programs are indexed to local prices. Federal Housing Administration loan insurance is guaranteed up to the level of local median home prices. Department of Housing and Urban Development (HUD) public housing and rental vouchers programs use local metropolitan-area income levels to determine eligibility, in combination with a local index of "fair market rents" to determine benefits. U.S. members of Congress have proposed but not passed legislation to index taxes and transfers to regional cost of living repeatedly: the Tax Equity Act, to index taxes; the Poverty Data Correction Act, to index the poverty line; and the COLA Fairness Act, to index social security payments.

 δpy), changes the expenditure function to $e[p, u, \tau(m - \delta py); Q] = \min_{x,y}[x + py + \tau(m - \delta py) : U(x, y; Q) \ge u]$. Differentiating the mobility condition and using the envelope theorem yields the log-linearized mobility condition

$$\hat{Q}^{j} = (1 - \delta \tau') s_{v} \hat{p}^{j} - (1 - \tau') s_{w} \hat{w}^{j}, \qquad (14)$$

which replaces (4a). Solving, the tax differential with the deduction also depends negatively on the home-good price differential, providing a level of cost indexation proportional to δ :

$$\frac{d\tau^{j}}{m} = \tau' \left(s_{w} \hat{w}^{j} - \delta s_{y} \hat{p}^{j} \right)$$

$$= \tau' \frac{s_{w} \hat{w}_{0}^{j} - \delta s_{y} \hat{p}_{0}^{j}}{1 - \tau' \left(\theta_{L} / \theta_{N} \right) \left(s_{w} / s_{R} \right) - \delta \tau' \left(s_{y} / s_{R} \right) \left[\phi_{L} - \phi_{N} \left(\theta_{L} / \theta_{N} \right) \right]}.$$
(15)

The second equality relates the tax differential to the pretax differentials in (6) and (9b) and results from subtracting $\hat{Q}^j = s_w \hat{p}_0^j - s_w \hat{w}_0^j$ from (14), substituting in (5) and (8b), and rearranging. The denominator reflects two multiplier effects: heavily taxed cities see wages rise and homegood prices fall, raising taxes through both higher wages and smaller deductions.

When $\delta < 1$, workers in cities with high trade productivity or low home productivity still pay higher taxes because the primary wage-tax effect is larger than the cost-indexation effect from the deduction. Taxes fall more precipitously with quality-of-life advantages, as the higher cost of living from quality of life is partly offset through the deduction.¹²

C. Including State Tax Differences

Differences in within-state tax burdens are worth considering, as wages and prices often vary significantly within a state, whereas state services largely do not. State tax differentials are computed by multiplying state tax and deduction rates by the wage and price differentials within state,

$$\frac{d\tau_{s}^{j}}{m} = \tau_{s}^{\prime} [s_{w}(\hat{w}^{j} - \hat{w}^{s}) - \delta_{s} s_{y}(\hat{p}^{j} - \hat{p}^{s})], \qquad (16)$$

where τ_s' and δ_s are marginal tax and deduction rates at the state level,

¹² This can be seen by substituting (6) and (9b) into

$$d\tau^{j}/m = \tau'\{(1-\delta)s_{j}s_{w}(s_{w}\dot{A}_{X}^{j}-s_{w}\dot{A}_{Y}^{j}) - [\delta s_{j}(\phi_{L}\theta_{N}-\phi_{N}\theta_{L}) + \theta_{L}s_{w}]\dot{Q}^{j}\}/[\theta_{N}s_{R}-\tau'\theta_{L}s_{w}-\delta\tau's_{j}(\phi_{L}\theta_{N}-\phi_{N}\theta_{L})].$$

The effect of federal taxes on prices or employment with cost indexation or deductions is determined by substituting $d\tau^{j}/m$ from (13) or (15) into eqq. (5), (8a), (8b), and (10).

net of federal deductions, and \hat{w}^s and \hat{p}^s are the differentials for state *S* as a whole, relative to the entire country. These state tax rates should incorporate sales as well as income taxes, since sales taxes reduce the buying power of labor income. The total tax differential for a city is the sum of the federal tax differential and the state tax differential.

VI. Simulation of Tax Differences across the United States

The theoretical model above may be used to simulate the effects of differential federal taxation on prices, employment, and welfare across the United States. This requires calibrating the economic parameters of the model and estimating wage, housing cost, federal spending, and quality-of-life differentials across metropolitan areas.

A. Calibrating the Model

An overview of the calibration is presented here. Alternative calibrations are considered in several sensitivity checks. Given that parameters are known with limited certainty, I use round fractions for ease.

Looking first at income shares, labor, s_w , receives 75 percent of income; capital, s_I , 15 percent; and land, s_R , 10 percent. Housing cost differences are used to measure home-good price differences. Using this measure requires that the expenditure share for home goods equals the expenditure share on housing of 22 percent plus the estimated expenditure share on nonhousing home goods of 14 percent, to produce $s_v = 0.36$; see Albouy (2008*a*) for details. From national accounts, the government expenditure share, s_T , is 15 percent. The cost shares depend on a number of sources. For traded goods, the cost share of land, θ_{I} , is 2.5 percent; the cost share of capital, θ_{K} , is 15 percent; and the cost share of labor, θ_N , is 82.5 percent. For home goods, the cost share of land, ϕ_L , is 23 percent; the cost share of capital, ϕ_K , is 15 percent; and the cost share of labor, ϕ_N is 62 percent. The cost and expenditure shares are consistent with the income shares and imply that the ratio λ_I / λ_N , which determines the fraction of taxes capitalized into wages, is equal to 23 percent.

The elasticity of employment with respect to local taxes, ε , is taken at -6.0 based on two methods, each yielding similar estimates. The first is to use direct reduced-form estimates of ε from Bartik's (1991) metaanalysis of the effect of local taxes on local levels of output and employment, controlling for local public spending. The second is to infer ε by directly calibrating a derived theoretical equation for employment changes using the above parameters as well as elasticities of substitution taken from the literature.

The marginal federal income tax rate on gross wages, τ' , of 33.3

percent is equal to the average marginal tax rate from TAXSIM (Feenberg and Coutts 1993) of 25.1 percent plus the marginal payroll tax rate on both the employer and employee sides, net of additional social security benefits (Boskin et al. 1987) of 8.2 percent. The federal deduction level, δ , is set at 0.257, which is far less than one because of renters, nonitemizing owners, nonhousing home goods, and the inability to deduct from payroll taxes.¹³

At the state level, the effective marginal tax rate on wages is 6.2 percentage points on average, from 0 points in Alaska to 8.8 percent in Minnesota. Wage differences within state are only 44 percent as large, on average, as wage differences within the entire country. Thus, total tax differences may be approximated by increasing the federal marginal tax rate by $6.2 \times 0.44 = 2.7$ points to 36 percent, although state tax differentials below are calculated exactly using equation (16).

B. Estimates of Wage, Price, and Spending Differentials

Wage and home-good price differentials are estimated using 5 percent samples of census data from the 2000 Integrated Public Use Microdata Series (IPUMS). Home-good price differentials are based on housing costs, as they are a prime determinant and predictor of cost-of-living differences. Cities are defined at the metropolitan statistical area (MSA) level using 1999 Office of Management and Budget (OMB) definitions. Consolidated MSAs are treated as a single city (e.g., San Francisco includes Oakland and San Jose), as are the nonmetropolitan areas of each state. This classification produces a total of 241 cities and 49 state-level collections of nonmetropolitan areas.

Interurban wage differentials, \hat{w}^{j} , are calculated from the logarithm of hourly wages for full-time workers, ages 25–55. These differentials control for skill differences across workers to provide an analogue to the representative worker in the model. Thus, log wages are regressed on city indicators, μ_{j}^{w} , and on extensive controls, X_{ij}^{w} —each fully interacted with gender—for education, experience, race, occupation, industry, and veteran, marital, and immigrant status, in an equation of the form $\ln w_{ij} = X_{ij}^{w}\beta^{w} + \mu_{j}^{w} + \varepsilon_{ij}^{w}$. The estimates of μ_{j}^{w} are used as the wage differential for city j and are interpreted as the causal effect of city j's attributes on a worker's wage. Identifying these differentials requires that workers do not sort across cities according to their unobserved skills. This assumption may not hold: Glaeser and Maré (2001)

¹³ Effects of a progressive tax system were also explored. A progressive tax schedule increases the variance of tax differentials, increasing the associated deadweight loss in (11). Because wage differentials are small relative to the tax schedule, they lead to only moderate changes in tax rates. A generous calculation produced at most a 5 percent increase in the deadweight burden calculation.

argue that up to one-third of the urban-rural wage gap could be due to selection, suggesting that at least two-thirds of wage differentials are valid, although this issue deserves greater investigation. At the same time, it is possible that the estimates could be too small, as some control variables, such as occupation or industry, could depend on where the worker locates.¹⁴

Housing values and gross rents reported in the census are used to calculate home-good price differentials, \hat{p}^{j} . To reduce measurement error from imperfect recall or rent control, the sample includes only units that were acquired in the last 10 years. Price differentials are calculated in a manner similar to wage differentials, using a regression of rents and values on flexible controls—interacted with tenure—for size, rooms, acreage, commercial use, kitchen and plumbing facilities, type and age of building, and the number of residents per room. Proper identification of housing-cost differences requires that average unobserved housing quality does not vary systematically across cities.¹⁵

Table 1 presents wage and housing-cost differentials in 2000 for selected metro areas and by census division and metropolitan size. Figure 3 graphs wage differentials against housing-cost differentials for all metro and nonmetro areas. Most large cities have above-average wages and housing costs; across cities of the same size, wages and costs tend to be higher in the Northeast and the Pacific. Overall, wages and housing costs are positively correlated, as reflected in the regression line.

Figure 3 plots a log-linearized mobility condition for cities with average quality of life, $\hat{Q}^j = 0$, and a log-linearized zero-profit condition for cities with average productivity in both sectors, $\hat{A}^j_X = \hat{A}^j_Y = 0$. Quality of life in a particular city is seen from how far its marker is to the right of this condition. Cities above the zero-profit condition have either high trade productivity or low home productivity, although without data on land rents, trade and home-productivity differences are not separately identified—nor do they need to be for this simulation. Quality-of-life and

¹⁴ Obviously, workers do not all have the same endowments and tastes or pay the same marginal tax rate, nor are they equally sensitive to productivity differences. However, as shown in Albouy (2008*b*), workers with different tastes and endowments can be aggregated without serious complications, as long as each is weighted by his or her share of income (which is done, although it has little impact on the estimates). Furthermore, many workers report receiving little income other than labor income. Yet, given the static nature of the model, a worker's choices should be modeled to account for a worker's permanent income, which includes a large nonlabor component, particularly if implicit rental earnings from one's own home are included.

¹⁵ Malpezzi, Chun, and Green (1998) determine that similar housing-cost indices derived from the census perform as well as or better than most other indices. Because home-good prices have only a minor effect on tax differentials, and as rent and housing-price differentials are highly correlated, the simulation is not very sensitive to how housing-cost differentials are estimated.

TABLE 1
Adjusted Wage, Housing-Cost, and Federal-Spending Differences
across Areas, 2000

	Population	Wage	Housing Cost	Federal Spending
	(1)	(2)	(3)	(4)
Metro area:				
San Francisco, CA	7,039,362	.26	.75	.011
New York, NY	21,199,865	.21	.42	003
Detroit, MI	5,456,428	.13	.09	009
Hartford, CT	1,183,110	.15	.15	.003
Chicago, IL	9,157,540	.14	.22	.001
Washington, DC	7,608,070	.13	.17	.006
Philadelphia, PA	6,188,463	.12	.07	.003
Boston, MA	5,819,100	.14	.35	.000
Minneapolis, MN	2,968,806	.09	.06	019
Los Angeles, CA	16,373,645	.13	.40	003
Jacksonville, FL	1,100,491	07	09	.006
Oklahoma City, OK	1,083,346	12	21	006
Norfolk, VA	1,569,541	11	07	013
Tucson, AZ	843,746	11	.00	.007
Killeen, TX	312,952	23	23	.025
Census division:				
Middle Atlantic	39,668,438	.08	.11	.000
Pacific	45,042,272	.10	.36	.001
New England	13,928,540	.07	.18	002
East North Central	45,145,135	.00	09	003
South Atlantic	51,778,682	03	06	001
West South Central	31,440,101	07	21	.001
Mountain	18,174,904	05	.02	.002
East South Central	17,019,738	12	30	.000
West North Central	19,224,096	11	25	.006
Metro population:				
Population >5 million	81,606,427	.16	.32	.000
Population 1.5–4.9 million	55,543,090	.03	.05	005
Population .5–1.4 million	40,499,870	03	07	.000
Population <.5 million	36,417,747	09	15	002
Nonmetro areas	67,354,772	14	28	.005
U.S. standard deviation		.13	.29	.011
U.S. mean absolute deviation		.11	.24	.008

NOTE.-Consolidated Metropolitan Statistical Areas (CMSA) used to define the largest metro areas. Wage, housing cost, and federal spending differentials are adjusted to control for observable characteristics.

productivity estimates across U.S. cities are reported and explained in Albouy (2008a, 2009).¹⁶

To investigate federal spending differentials, data are taken from the Consolidated Federal Funds Report (CFFR), available from the U.S. Census of Governments. Spending is divided into three categories: (i) government wages and contracts, (ii) benefits to nonworkers, and (iii)

¹⁶ Ignoring state taxes, the slope of the mobility condition is $s_y(1 - \delta \tau') / [s_w(1 - \tau')]$, and the slope of the zero-profit condition is $-\theta_L / (\theta_N \phi_L - \theta_L \phi_N)$. The capitalization of a quality-of-life improvement or a federal tax reduction (modeled as a head tax) on wages and housing prices is illustrated by shifting the mobility condition to the right. The capitalization of an increase in firm productivity or a decrease in home productivity is modeled by shifting the zero-profit condition to the right.



FIG. 3.—Wage levels and housing costs across areas, 2000

other spending. The first category consists of federal government purchases of goods and labor services; if these purchases are made at cost, they should not be considered transfers.¹⁷ The second category includes spending that benefits individuals who are typically inactive in the labor market, such as retirees and full-time students, including social security and Medicare. The remaining category of other spending is more likely to benefit workers according to their location: it includes most government grants, such as for welfare, Medicaid, infrastructure, and housing subsidies. Spending differentials are adjusted to control for a limited set of population characteristics in a city, such as average age and percent minority, to provide a spending differential applicable to a representative worker. The adjusted differentials for other spending are reported as a fraction of household income in table 1.

C. Tax Differences and Their Effects

Using the base calibration and estimates of \hat{w}^{j} and \hat{p}^{j} for 2000, table 2 reports estimates of tax differentials and their effects across selected cities, and by census division and metropolitan size.¹⁸ The wage and deduction components of the federal tax differential in (15) are in columns 1 and 2, with the sum in column 3. State tax differentials are in column 4, and the total combined tax differential is in column 5. A kernel density estimate of these total tax differentials is drawn in figure 4.

The unequal distribution of taxes is substantial: the mean absolute deviation of federal tax differentials equals 2.2 percent of income, and with state taxes this rises to 2.4 percent. Starting at an average federal tax rate of 17 percent, a worker moving from a typical low-wage city to a typical high-wage city sees her average tax rate rise from 14.8 percent to 19.2 percent, paying 27 percent more in federal taxes. Although tax differences are compensated for in local prices, this represents a horizontal transfer of \$269 billion (in 2008) from workers in high-wage areas to similarly skilled workers in low-wage areas.¹⁹

According to the simulation, the tax differential from equation (15) is given numerically by $d\tau^{j}/m = 0.271 \hat{w}^{j} - 0.035 \hat{p}^{j}$. Tax differences are driven largely by wage differences, although price differences have some effect. This phenomenon is illustrated in figure 3 by the upward-sloping

¹⁷ Weingast, Shepsle, and Johnsen (1981) explain when localized spending should be treated as a transfer.

¹⁸ A full list is provided in Albouy (2008b).

¹⁹ The average federal tax rate of 17 percent includes federal income taxes and payroll taxes, appropriately adjusted (Congressional Budget Office 2003). Multiplying the mean absolute deviation of federal tax differentials, 0.221, by personal income in 2008 of \$12.11 trillion produces a figure of \$269 billion. Using GDP produces \$317 billion, or adjusted gross income of roughly \$189 billion.

average iso-tax line: a city's federal tax differential is proportional to the vertical distance of the city marker above this line. Empirically, the deductions tend to reduce tax differences across areas, as wages and housing costs are positively related. Figure 4 shows how eliminating the deduction would change the distribution of federal taxes across cities, increasing the tax differential gradient by 34.5 percent. Thus, without the deduction, the average tax differential would be 3.2 percent, making the distribution of federal taxes even more unequal.²⁰

Each city's tax differential depends on its attributes, according to the numeric analogue of equation (7): $d\tau^{j}/m = -0.185\hat{Q}^{j} + 0.228\hat{A}_{X}^{i} - 0.025\hat{A}_{Y}^{j}$, which adjusts for deductions and state taxes. Thus, federal taxes depend quite positively on a city's trade productivity and negatively on quality of life and, to a lesser extent, home productivity. The mean absolute deviation of \hat{Q}^{j} is 0.037, whereas the corresponding figure for \hat{A}_{X}^{i} , assuming $\hat{A}_{Y}^{j} = 0$ for all *j*, is 0.111. Accordingly, the average tax differential from quality-of-life differences alone is only 0.7 percent, whereas the same from productivity differences. As a result, tax burdens are highest in large, productive cities in the Northeast, Midwest, and Pacific, whereas small, less productive towns and nonmetropolitan areas, particularly in the South, receive a large tax break.

The total tax differentials are considerable relative to typical differences in local taxes. Any local official would consider a permanent 3 percent tax on local residents without any compensating services to be a fiscal calamity. Yet, central governments are imposing this situation on cities such as Chicago, New York, and San Francisco. However, an unconditional grant of 3 percent of income in perpetuity dwarfs almost any pork-barrel project. Relative to the national average, this is what workers in cities such as Norfolk and Tucson, as well as most nonmetropolitan areas, effectively receive from the federal government.

These large tax differentials have considerable effects on prices and employment, seen in the last four columns of table 2. For example, the additional taxes paid to Washington and Albany by New York City raise wages by 1.3 percent, lower long-run housing costs by 9 percent, and lower land values by 35 percent. The employment effect is especially striking, stating that employment is 23 percent lower than in an undistorted equilibrium. This effect may seem too large, but it may be reasonable in the long run, as sizable federal taxes first affected average workers in World War II. The rise of the income tax is certainly consistent

²⁰ Since the existing tax system has a deduction, the tax differentials with no deduction in fig. 4 and col. 6 of table 3 are based on the counterfactual wage without a deduction. Counterfactual wages are simply $\hat{w}_0^i = \hat{w}^j - d\hat{w}^j$. The tax differentials with no deduction are similar to the ones for which the deduction is just ignored and \hat{w}^j is used.

				TA	BLE 2					
	TA3	(DIFFEREN	TIALS AND	Their Effi	ECTS ON PRICE	s and Employm	ENT, 2000			
		FEDERA	l Tax Diffei	JENTIAL			Tor	tal Tax Dii	fferential Ef	FECTS
	Tax Payment Rank	From Wage (1)	From Deduct (2)	Total Federal (3)	STATE TAX Differential (4)	TOTAL TAX Differential (5)	Wage (6)	Housing Cost (7)	Land Rent (8)	Employment (9)
Metro area:						n. F				
San Francisco, CA	1	.068	023	.045	.004	.048	.015	103	480	288
New York, NY	2	.054	013	.041	.002	.043	.013	093	434	261
Detroit, MI	6	.035	003	.032	.004	.036	.011	076	355	213
Hartford, CT	4	.039	005	.035	.001	.035	.011	076	352	211
Chicago, IL	9	.035	007	.029	.003	.031	600.	067	310	186
Washington, DC	7	.034	005	.029	.002	.031	600.	066	306	183
Philadelphia, PA	8	.030	002	.028	.002	.030	600.	065	303	182
Boston, MA	6	.035	011	.025	100.	.026	.008	056	259	155
Minneapolis, MN	10	.023	002	.021	.005	.026	.008	055	256	154
Los Angeles, CA	15	.033	012	.021	000.	.021	.006	045	209	126
Jacksonville, FL	114	019	.003	016	000.	015	005	.033	.153	.092
Oklahoma City, OK	160	032	.006	026	.002	024	007	.051	.239	.143
Norfolk, VA	188	028	.002	026	004	030	009	.064	.300	.180
Tucson, AZ	195	029	.000	029	003	032	010	.069	.322	.193
Killeen, TX	241	060	.007	053	005	058	018	.125	.582	.349

Census division:										
Middle Atlantic	-	.021	003	.017	000.	.017	.005	037	172	103
Pacific	5	.026	011	.015	000.	.015	.004	031	145	087
New England	3	.017	006	.012	000.	.012	.004	025	117	070
East North Central	4	.001	.003	.003	000.	.003	100.	007	033	020
South Atlantic	5	008	.002	006	000.	006	002	.013	.062	.037
West South Central	9	019	.006	012	000.	012	004	.026	.123	.074
Mountain	7	013	001	013	000.	013	004	.029	.135	.081
East South Central	×	030	600.	021	000.	021	006	.044	.207	.124
West North Central	6	029	.008	021	000.	021	006	.046	.213	.128
Metro population:										
Population >5 million	-	.041	010	.031	.002	.033	.010	071	330	198
Population 1.5-4.9 million	5	.007	001	.006	.001	.007	.002	015	071	043
Population .5–1.4 million	3	008	.002	006	000.	005	002	.011	.053	.032
Population <.5 million	4	022	.005	018	002	019	006	.041	.193	.116
Nonmetro areas	5	036	600.	027	002	030	009	.064	.296	.178
U.S. standard deviation		.034	600.	.026	.003	.028	.008	.060	.278	.167
U.S. mean absolute deviation		.029	200.	.022	.002	.024	200.	160.	.239	.143
NorrThe federal tax differentials 1 of table 3.	are calcula	ted from eq. (19	5), and the stat	e tax differentia	ls from eq. (16).	Total tax different	ial effects are ba	ased on the ma	in calibration p	resented in col.



FIG. 4.—Differential tax burdens with and without deduction

with the migration of people and jobs over the last 60 years from the high-wage "rust belt" to the low-wage "sun belt" (Kim and Margo 2004).

The nationwide effects on prices, employment, and welfare for a number of different calibrations are given in table 3. The economic and tax parameters of these calibrations are displayed in the first panel, followed by the mean absolute deviations in outcomes and the deadweight loss of taxation throughout the economy. All effects are averaged using the total population size of each area as weights.

The benchmark case, shown in column 1, reveals the overall significance of differential federal taxation nationwide. In a typical high-wage city, workers pay 2.4 percent more of their income in taxes, which causes land values to be 21 percent lower. Workers are compensated for the tax differential through a 0.7 percent increase in wages, increasing their pretax incomes by slightly over 0.5 percent, and a 5.0 percent reduction

SIMULATED F	FFECTS OF TAX	DIFFERENTIA	LS ACROSS AI	le Areas for A	LTERNATE CALI	(BRATIONS		
		All Land in		Larger Emplovment	Wage Differentials	Housing	Federal Taxes Only.	Adding Federal
	Benchmark	Home	Smaller	Response:	Two-Thirds	Deductions	No State	Spending
	Case (1)	Goods (2)	Land Share (3)	Cobb-Douglas	Estimated Size	Eliminated (6)	Taxes	Differences (8)
Economic parameters:				*	n A			
Home-goods share, s_{*}	.360	.360	.360	.360	.360	.360	.360	.360
Traded-good land share, θ_L	.025	000.	.013	.025	.025	.025	.025	.025
Traded-good labor share, θ_N	.825	.850	.825	.825	.825	.825	.825	.825
Home-good land share, ϕ_L	.233	.278	.117	.233	.233	.233	.233	.233
Home-good labor share, ϕ_N	.617	.572	.617	.617	.617	.617	.617	.617
Elasticity of employment to tax/income, ε	-6.000	-6.000	-6.000	-9.370	-6.000	-6.000	-6.000	-6.000
Implied share of income to land, s_R	.100	.100	.050	.100	.100	.100	.100	.100
Tax parameters:								
Marginal tax rate, τ'	.361	.361	.361	.361	.361	.361	.333	.361
Deduction level, δ	.291	.291	.291	.291	.291	.000	.257	.291
Average percent effects (mean abs. dev.):								
Tax differential, $E d\tau^j $.024	.024	.024	.024	.014	.032	.022	.027
Wage effect, $E d\hat{w}^{j} $.007	.000	.007	.007	.004	.010	.007	.008
Home-good price effect, $E d\hat{p}^{j} $.051	.066	.051	.051	.029	.069	.048	.059
Land rent effect, $E d\hat{n}' $.239	.239	.478	.239	.136	.323	.222	.274
Employment effect, $E d \hat{N}^j $.143	.143	.143	.224	.082	.194	.133	.164
Deadweight loss from locational inefficiency:								
As a percent of income, $DWL/(mN^{TOT})$.23%	.23%	.23%	.36%	.07%	.43%	.20%	.31%
Total, billions per year, 2008 dollars	28	28	28	44	6	52	24	37
Per capita per year, 2008 dollars	93	93	93	146	30	174	82	124
NOTE.—Wage, land rent, home-good price, and em on a personal income measure of \$12.11 billion. The	ployment effects are Cobb-Douglas elastici	calculated from ity of employmen	eqq. (5), (8a), (8l) at is derived in or	o), and (10). Deadw nine Apps. A and B.	eight loss is determi. See text for further	ned from eq. (11) : detail.), and the dollar	values are based

TABLE 3 Effects of Tax Differentiats across All Areas for Altery in the housing prices, reflecting a cost-of-living reduction of 1.8 percent. Thus, workers are compensated for higher taxes more through lower costs than through higher wages.

The negative employment effect on a typical high-wage city is 14 percent. Taken together, the employment effects create a substantial deadweight loss of about 0.23 percent of income a year, or \$28 billion in 2008. As these numbers are based on a calibrated model, they should not be taken as absolute truth, but they do provide a sense of the magnitude of the impacts and costs caused by the unequal distribution of federal taxes.²¹

Alternative calibrations in table 3 are shown in columns to the right. In column 2, all land is devoted to home-good production, keeping the total share of income to land constant: in this case, wage differentials are unaffected by taxes, whereas home-good price differentials are affected more. In column 3, the cost shares of land in both sectors are reduced by one-half, with mobile capital taking up the remaining costs; this doubles the impact on land rents without changing any of the other quantities.

Column 4 shows that if ε is -9.37, which corresponds to when production and preferences are Cobb-Douglas, the employment effects and deadweight loss are increased proportionally. Column 5 cuts wage differentials down to two-thirds their original size, in case unobserved selection makes the estimated differentials too large: this lowers the differential taxes and price and employment effects by 41 percent and reduces the deadweight loss to only 0.07 percent of income. Column 6 reveals that if the deduction is eliminated but tax rates on labor are held constant, then the tax effects would become 35 percent larger and the deadweight loss would increase to 0.43 percent of income. Finally, column 7 looks at the effect of federal taxes only, ignoring state taxes. Since federal taxes account for 92 percent of tax differences, the effects are only slightly smaller.

D. The Distribution of Federal Spending

The unequal burden of federal taxation would be much less of an issue if it was compensated for by federal spending differences. To explore this possibility, table 4 reports coefficients from regressions of spending differentials, both raw and adjusted, on tax differentials in 2000. In the raw differentials, there is a positive correlation with federal purchases

²¹ In the base calibration, agglomeration effects could dampen the positive effect of taxes on wages. According to Rosenthal and Strange (2004), the elasticity of wages with respect to population size due to agglomeration is close to 4 percent. At this level, a 17 percent reduction in employment from taxes reduces wages by 0.7 percent, which would offset the 0.7 percent predicted increase in wages due to higher land-to-labor ratios.

TABLE 4
DIFFERENTIAL FEDERAL SPENDING PATTERNS RELATIVE TO DIFFERENTIAL
TAXATION PATTERNS, 2000

Type of Federal Spending	All Spending (1)	Wages and Contracts (2)	Nonworker Benefits (3)	All Other Spending (4)
A. Raw spending differentials:				
federal tax differential	134	.305	261	025
	(.212)	(.158)	(.092)	(.070)
B. Adjusted spending differentials:				
federal tax differential	246	094	032	098
	(.157)	(.095)	(.025)	(.037)

NOTE. – Each entry corresponds to a separate regression, reporting the coefficient on the federal tax differential variable using alternate measures of federal spending as the dependent variable. Regressions weighted by population for all 290 observations. Robust standard are errors reported in parentheses. Adjusted spending differentials control for socioeconomic disparities as described in online App. C.

(wages and contracts), a negative correlation with nonworker benefits, and no correlation with other spending, the category closest to a locational transfer. Once population characteristics are controlled for, correlations for wages and contracts and nonworker benefits become negative and insignificant, while other spending, as well as aggregate spending, becomes negatively correlated with federal tax differentials. Although the federal government makes greater purchases in areas with higher wages, this arises from its need to purchase skilled labor.

Figure 5, which graphs "other spending" differentials against tax differentials, makes it clear that federal spending does not offset differences in federal taxation and may in fact do the opposite. Column 8 of table 3 simulates the effects of tax differentials net of other spending: these differentials have slightly larger variance, increasing the tax effects and deadweight loss by a small amount.

VII. Conclusion

Any tax on labor income creates an incentive for workers to leave highwage areas in favor of low-wage areas. Although mobile workers should be compensated for the resulting tax differences through adjustments in local prices and wages, the resulting geographic distribution of employment will be distorted, causing a substantial welfare loss.

The simulated effects of federal taxes on prices, employment, and welfare are based on the assumption that city attributes are unaffected by population movements. When city attributes are affected by population size, these effects could be smaller or larger than predicted. Furthermore, the distribution of city sizes may no longer be optimal even in the absence of federal taxes, which could ameliorate or aggravate preexisting distortions. Given the complexities of dealing with endog-



FIG. 5.—Federal spending and tax differentials across areas

enous attributes, these issues are left for further work (see Albouy and Seegert 2009).

Politicians who represent high-wage areas may legitimately complain that their districts pay a disproportionate share of federal taxes. However, in most countries, reforms to equalize the federal tax burden across areas would likely meet fierce political opposition. In the United States, highly taxed areas tend to be in large cities inside of populous states, which have low congressional representation per capita, making the prospect of reform daunting. In other countries, such as Canada, rural areas also receive disproportionate representation in national legislatures. Nevertheless, when considering federal tax reforms, policy makers should be aware of their spatial consequences on local prices, employment, and welfare.

References

- Albouy, David. 2008a. "Are Big Cities Really Bad Places to Live? Improving Quality-of-Life Estimates across Cities." Working Paper no. 14472 (November), NBER, Cambridge, MA.
- ——. 2008*b.* "The Unequal Geographic Burden of Federal Taxation." Working Paper no. 13995 (May), NBER, Cambridge, MA.
- ———. 2009. "What Are Cities Worth? Land Rents, Local Productivity, and the Capitalization of Amenity Values." Working Paper no. 14981 (May), NBER, Cambridge, MA.
- Albouy, David, and Nathan Seegert. 2009. "The Optimal Distribution of Population across Cities and the Private-Social Wedge." Manuscript, Univ. Michigan.
- Antras, Pol. 2004. "Is the U.S. Aggregate Production Function Cobb-Douglas? New Estimates of the Elasticity of Substitution." *Contributions to Macroeconomics* 4 (1): 1–33.
- Bartik, Timothy J. 1991. Who Benefits from State and Local Economic Development Policies? Kalamazoo, MI: Upjohn Inst.
- Beeson, Patricia E., and Randall W. Eberts. 1989. "Identifying Productivity and Amenity Effects in Interurban Wage Differentials." *Rev. Econ. and Statis.* 71 (August): 443–52.
- Berndt, Ernst. 1976. "Reconciling Alternative Estimates of the Elasticity of Substitution." *Rev. Econ. and Statis.* 58 (February): 59–68.
- Blanchard, Olivier-Jean, and Lawrence F. Katz. 1992. "Regional Evolutions." Brookings Papers Econ. Activity (1): 1–61.
- Boskin, Michael J., Laurence J. Kotlikoff, Douglas J. Puffert, and John B. Shoven. 1987. "Social Security: A Financial Appraisal across and within Generations." *Nat. Tax J.* 40 (March): 19–34.
- Brady, Peter, Julie-Anne Cronin, and Scott Houser. 2003. "Regional Differences in the Utilization of the Mortgage Interest Deduction." *Public Finance Rev.* 31 (July): 327–66.
- Bureau of Labor Statistics. 2002. *Consumer Expenditures in 2000*. Washington, DC: Government Printing Office.
- Cassimatis, Peter J. 1969. *Economics of the Construction Industry*. New York: National Indus. Conf. Bd.
- Congressional Budget Office. 2003. *Effective Federal Tax Rates*, 1997 to 2000. Washington, DC: Government Printing Office.
- Davis, Morris A., and Jonathan Heathcoate. 2007. "The Price and Quantity of Residential Land in the United States." *J. Monetary Econ.* 54 (November): 2595–2620.
- Dubay, Curtis S. 2006. "Federal Tax Burdens and Expenditures by State: Which States Gain Most from Federal Fiscal Operations." Special Report no. 139, Tax Found., Washington, DC.
- Feenberg, Daniel, and Elisabeth Coutts. 1993. "An Introduction to the TAXSIM Model." J. Policy Analysis and Management 12 (Winter): 189–94.
- Feenberg, Daniel R., Andrew W. Mitrusi, and James M. Poterba. 1997. "Distributional Effects of Adopting a National Retail Sales Tax." Working Paper no. 5885 (November), NBER, Cambridge, MA.
- Gladwell, Malcolm. 1996. "U.S. to New York: It's Still Dutch Treat." Washington Post, March 7.
- Glaeser, Edward L. 1998. "Should Transfer Payments Be Indexed to Local Price Levels?" *Regional Sci. and Urban Econ.* 28 (January): 1–20.

- Glaeser, Edward L., and David Maré. 2001. "Cities and Skills." J. Labor Econ. 19 (April): 316–42.
- Glaeser, Edward L., and Jesse M. Shapiro. 2003. "The Benefits of the Home Mortgage Interest Deduction." In *Tax Policy and the Economy*, vol. 17, edited by James M. Poterba. Cambridge, MA: MIT Press.
- Goodman, Allen C. 1988. "An Econometric Model of Housing Price, Permanent Income, Tenure Choice, and Housing Demand." J. Urban Econ. 23 (May): 327– 53.
- Gyourko, Joseph, and Todd Sinai. 2003. "The Spatial Distribution of Housing-Related Ordinary Tax Benefits." *Real Estate Econ.* 31 (4): 527–75.
- ——. 2004. "The (Un)Changing Geographical Distribution of Housing Tax Benefits: 1980 to 2000." In *Tax Policy and the Economy*, vol. 18, edited by James M. Poterba. Cambridge, MA: MIT Press.
- Gyourko, Joseph, and Joseph Tracy. 1989. "The Importance of Local Fiscal Conditions in Analyzing Local Labor Markets." *J.P.E.* 97 (October): 1208–31.
- _____. 1991. "The Structure of Local Public Finance and the Quality of Life." *J.P.E.* 99 (August): 774–806.
- Harberger, Arnold C. 1962. "The Incidence of the Corporate Income Tax." *J.P.E.* 70 (1962): 215–40.
- . 1964. "Principles of Efficiency: The Measurement of Waste." A.E.R. Papers and Proc. 54 (May): 58–76.
- Hoffman, David K., and J. Scott Moody. 2003. "Federal Income Taxes and the Cost of Living: States with Lower Cost of Living Enjoy Great Federal Tax Advantage." Special Report no. 125, Tax Found., Washington, DC.
- Ioannides, Yannis M., and Jeffrey E. Zabel. 2003. "Neighborhood Effects and Housing Demand." J. Appl. Econometrics 18 (September): 563-84.
- Jones, Ronald W. 1965. "The Structure of Simple General Equilibrium Models." *J.P.E.* 73 (December): 557–72.
- Kaplow, Louis. 1995. "Regional Cost-of-Living Adjustments in Tax/Transfer Schemes." Tax Law Rev. 51 (Winter): 175–98.
- ——. 1996. "Fiscal Federalism and the Deductibility of State and Local Taxes under the Federal Income Tax." *Virginia Law Rev.* 82 (April): 413–92.
- Keiper, Joseph S., Ernest Kurnow, Clifford D. Clark, and Harvey H. Segal. 1961. *Theory and Measurement of Rent.* Philadelphia: Chilton.
- Kim, Sukkoo, and Robert Margo. 2004. "Historical Perspectives on U.S. Economic Geography." In *Handbook of Regional and Urban Economics*, vol. 4, edited by J. V. Henderson and J.-F. Thisse. Amsterdam: North-Holland.
- Knoll, Michael S., and Thomas D. Griffith. 2003. "Taxing Sunny Days: Adjusting Taxes for Regional Living Costs and Amenities." *Harvard Law Rev.* 116 (February): 987–1023.
- Kotlikoff, Laurence J., and Lawrence H. Summers. 1987. "Tax Incidence." In *Handbook of Public Economics*, vol. 2, edited by A. J. Auerbach and M. Feldstein. Amsterdam: North-Holland.
- Krueger, Alan B. 1999. "Measuring Labor's Share." A.E.R. Papers and Proc. 89 (May): 45–51.
- Lucas, Robert E. 1969. "Labor-Capital Substitution in U.S. Manufacturing." In *The Taxation of Income from Capital*, edited by Arnold Harberger and Martin J. Bailey. Washington, DC: Brookings Inst.
- Malpezzi, Stephen, Gregory H. Chun, and Richard K. Green. 1998. "New Placeto-Place Housing Price Indexes for U.S. Metropolitan Areas, and Their Determinants." *Real Estate Econ.* 26 (June): 235–74.

- McDonald, John F. 1981. "Capital-Land Substitution in Urban Housing: A Survey of Empirical Estimates." *J. Urban Econ.* 9 (March): 190–211. Mieszkowski, Peter. 1972. "The Property Tax: An Excise Tax or a Profits Tax?"
- J. Public Econ. 1 (April): 73-96.
- Moretti, Enrico. 2008. "Real Wage Inequality." Working Paper no. 14370 (November), NBER, Cambridge, MA.
- Moynihan, Daniel Patrick. 2000. "Introduction." In The Federal Budget and the States: Fiscal Year 1999. Washington, DC: Taubman Center for State and Local Government, John F. Kennedy School of Government, U.S. Senate Printing Office
- Phillips, Joseph M., and Ernest P. Goss. 1995. "The Effect of State and Local Taxes on Economic Development: A Meta-analysis." Southern Econ. J. 62 (October): 320-33.
- Poterba, James M. 1992. "Taxation and Housing: Old Questions, New Answers." A.E.R. Papers and Proc. 82 (May): 237-42.
- -. 1998. "The Rate of Return to Corporate Capital and Factor Shares: New Estimates Using Revised National Income Accounts and Capital Stock Data." Carnegie-Rochester Conf. Series Public Policy 48:211-46.
- Rappaport, Jordan. 2008. "A Productivity Model of City Crowdedness." J. Urban Econ. 63 (March): 715-22.
- Roback, Jennifer. 1980. "The Value of Local Urban Amenities: Theory and Measurement." PhD diss., Univ. Rochester.
- . 1982. "Wages, Rents, and the Quality of Life." J.P.E. 90 (December): 1257-78.
- Rosen, Harvey. 1985. "Housing Subsidies: Effects on Housing Decisions, Efficiency, and Equity." In Handbook of Public Economics, vol. 1, edited by M. Feldstein and A. Auerbach. Amsterdam: North-Holland.
- Rosen, Sherwin. 1979. "Wages-Based Indexes of Urban Quality of Life." In Current Issues in Urban Economics, edited by P. Mieszkowski and M. Straszheim. Baltimore: Johns Hopkins Univ. Press.
- Rosenthal, Stuart, and William Strange. 2004. "Evidence on the Nature and Sources of Agglomeration Economies." In Handbook of Regional and Urban Economics, vol. 4, edited by J. V. Henderson and J.-F. Thisse. Amsterdam: North-Holland.
- Ruggles, Steven, et al. 2004. Integrated Public Use Microdata Series: Version 3.0. Minneapolis: Minnesota Population Center.
- Shapiro, Jesse M. 2006. "Smart Cities: Quality of Life, Productivity, and the Growth Effects of Human Capital." Rev. Econ. and Statis. 88 (May): 324-35.
- Sieg, Holger, Dennis Epple, and Brett Gordon. Forthcoming. "A New Approach to Estimating the Production Function for Housing." A.E.R.
- Thorsnes, Paul. 1997. "Consistent Estimates of the Elasticity of Substitution between Land and Non-land Inputs in the Production of Housing." J. Urban Econ. 42 (July): 98-108.
- Tiebout, Charles M. 1956. "A Pure Theory of Local Expenditures." J.P.E. 64 (October): 416-24.
- Valentinyi, Ákos, and Berthold Herrendorf. 2008. "Measuring Factor Income Shares at the Sectoral Level." Rev. Econ. Dynamics 11 (October): 820-35.
- Weingast, Barry R., Kenneth A. Shepsle, and Christopher Johnsen. 1981. "The Political Economy of Benefits and Costs: A Neoclassical Approach to Distributive Politics." J.P.E. 89 (August): 642-64.
- Wildasin, David E. 1980. "Locational Efficiency in a Federal System." Regional Sci. and Urban Econ. 10 (November): 453-71.